Author's response to reviews

Title: Trends in complementary/alternative medicine use by breast cancer survivors: Comparing survey data from 1998 and 2005

Authors:

Heather S Boon (heather.boon@utoronto.ca)
Folashade Olatunde (shade.olatunde@utoronto.ca)
Suzanna M Zick (szick@med.umich.edu)

Version: 4 Date: 29 March 2007

Author's response to reviews:

Formatting changes requested
-------------------------------
The following changes were made:

Used numbers to denote affiliation, not letters.

Moved the email list to beneath the institution details.

Removed the 'Purpose' heading from the abstract.

Changes Author's contributions: -> Authors' contributions:

Aligned the title, and the references heading to the left.